Author's response to reviews

Title: Knowledge of vitamin D and perceptions and attitudes towards sunlight among Chinese middle-aged and elderly women: a population survey in Hong Kong

Authors:

Annie WC Kung (awckung@hkucc.hku.hk)
Kar-Kui Lee (kklee@graduate.hku.hk)

Version: 5 Date: 4 August 2006

Author's response to reviews:

Page 11, Discussion 2nd last sentence is changed to read "SPF>15 was chosen because sunscreen products with SPF<15 are not commercially available in Hong Kong."